Author's response to reviews

Title: Pre-pregnancy care for women with pre-gestational diabetes mellitus: a systematic review and meta-analysis

Authors:

Hayfaa A Wahabi (umlena@yahoo.com)
Rasmieh A Alzeidan (ras_zeidan@hotmail.com)
Samia A Esmaeil (drsoyo@hotmail.com)

Version: 3 Date: 2 September 2012

Author's response to reviews:

Dear Editor

We would like to thank the reviewer for her constructive comments on the 7th of August 2012 and below are point-to-point reply to those comments (All text changes are highlighted in yellow)

1. We removed the repeated phrase from the abstract.
2. Discussion:
   a) The sentence including (the CM) has been changed to (the rate of CM)
   b) The percentage sign (%) had been added to the HbA1C on page 9.
   c) The word “and” has been removed from page 9 as advised by the reviewer.
   d) The year of publication of the two studies has been added to the text.
3. Table 5 brackets were amended to be similar all through the table.
4. Figure 4&5 were improved.
5. The phrase (preconception care) was removed from Figure 5.

Best Regards
Dr Hayfaa A Wahabi